Author’s response to reviews

Title: Effect of green tea on the gastrointestinal absorption of amoxicillin in rats

Authors:

Tivadar Kiss (kiss.tivadar@pharmacognosy.hu)
Zoltán Tímár (timar@solvoc.om)
Andrea Szabó (szabo.andrea@med.u-szeged.hu)
Anita Lukács (lukacs.anita@med.u-szeged.hu)
Viktória Velky (velky@solvo.com)
Gábor Oszlánczi (octbintw@gmail.com)
Edina Horváth (korosine.edina@med.u-szeged.hu)
István Takács (taki.biotech@gmail.com)
István Zupkó (zupko@pharm.u-szeged.hu)
Dezső Csupor (csupor.dezso@pharmacognosy.hu)

Version: 3 Date: 29 Jun 2019

Author’s response to reviews:

Dear Elena Enioutina,

the presented AUC values they were calculated by means of GraphPad Prism. Concerning the unit, we are convinced that any kind of time*concentration unit is usable. However, in order to make our manuscript appropriate for the style of the Journal all the determined concentrations have been converted into microg/mL values and therefore the unit of the AUC has been also changed according to the editorial suggestion. Fig 1 and 2 are now updated.

Best regards, Dezső Csupor